Author’s response to reviews

Title: The role of T2* Gradient Echo in the diagnosis of tumefactive intrahepatic extramedullary hematopoiesis in myelodysplastic syndrome and diffuse hepatic iron-overload: A case report and literature review.

Authors:

abel belay (abelassefamd@gmail.com)
Andrew Bellizzi (andrew-bellizzi@uiowa.edu)
Alan Stolpen (alan-stolpen@uiowa.edu)

Version: 1 Date: 22 Sep 2017

Author’s response to reviews:

First reviewer

1) Add peripheral blood smear showing monocytosis and supporting the diagnosis of CMML-2
2) And a photo of CT angiogram demonstrating a mass of EMH
3) BCR-ABL fusion gene should be written as BCR-ABL1
4) When talking about MDS subtypes please indicate that you are using WHO classification.

Response: CT angiogram demonstrating a mass of EMH added, BCR-ABL fusion changed to BCR-ABL1. I have also indicated I'm using WHO classification for MDS subtypes. Picture of peripheral blood smear showing CMML has not been included, however, currently trying to track down the image.

Second reviewer

The case it self is interesting but need to be rewritten and focused on main topic

Response: Non-specific comment, however, paragraphs 1, 4 and 7 under discussion have been modified to focus on main topic